Author's response to reviews

Title: Healthcare professionals' intentions to use wiki-based reminders to promote best practices in trauma care: a survey protocol

Authors:

Patrick M Archambault (patrick.m.archambault@gmail.com)
France Légaré (france.legare@mfa.ulaval.ca)
André Lavoie (andre.lavoie.trauma.cha@ssss.gouv.qc.ca)
Marie-Pierre Gagnon (marie-pierre.gagnon@fsi.ulaval.ca)
Jean Lapointe (lcoutjla@videotron.ca)
Sylvie St-Jacques (sylvie.st-jacques@crsfa.ulaval.ca)
Julien Poitras (julienpoitras@videotron.ca)
Karine Aubin (Karine_Aubin@ssss.gouv.qc.ca)
Sylvain Croteau (sylvain.croteau@gmail.com)
Martin Pham-Dinh (martinphamdinh@gmail.com)

Version: 5 Date: 2 June 2010

Author's response to reviews: see over
Wednesday, June 2, 2010

Dear Drs Eccles and Mittman,

I am submitting a second revised version of the study protocol entitled: *Healthcare professionals’ intentions to use wiki-based reminders to promote best practices in trauma care: A survey protocol.*

Following my proof reading of the final documents, I have made the following minor changes to the main manuscript, figure 1 and figure 2.

Main manuscript:

1) In the final paragraph of the ‘Potential study limitations and how they will be addressed’ section, I changed ‘to consult the wiki’ to ‘to use the wiki’ (last sentence of the paragraph).

2) In the ‘Ethical aspects’ section, I changed the first sentence and now reads as follows: ‘This study protocol has been approved by the ethics review boards of all three hospitals in the study.’

In Figure 1, the pdf version that you sent me had a small mistake in that the arrow connecting the ‘perceived behavior control’ box to the ‘intention’ box overlapped in the ‘perceived behavior control box’. I am submitting a corrected version.

In Figure 2, I omitted to remove the names of the ‘CHAU HDL’ and ‘CHAUQ’ in the box beside the ‘phase 3’ box. I simply removed any mention of the names of these two hospitals to preserve anonymity.

Sincerely,

Patrick Archambault, M.D., MSc., FRCPC
Corresponding author
Emergency and Critical Care Medicine Specialist
Centre Hospitalier Affilié Universitaire Hôtel-Dieu de Lévis
Post-doctoral Fellow
Research center of the Centre hospitalier universitaire de Québec